Author's response to reviews

Title: Primitive Neuroectodermal Tumor (PNET) of the kidney: case report and carcinogenic hypothesis.

Authors:

Dr Giorgio Pomara (g.pomara@libero.it)
Francesco Cappello (francapp@hotmail.com)
Maria G. Cuttano (mg.cuttano@med.unipi.it)
Francesca Rappa (francapp@hotmail.com)
Dr Girolamo Morelli (g.pomara@libero.it)
Dr Pierantonio Mancini (g.pomara@libero.it)
Cesare Selli (c.selli@dc.med.unipi.it)

Version: 2 Date: 19 Sep 2003

Dear Sirs,

as you suggested We modified the abstract of our manuscript according to the instructions for authors. Moreover We added in the Acknowledgements section of the paper the text concerning patient's written consent.

Finally we would suggest the fifth and last reviewer: Prof. Linehan W. Marston. Urology Branch-NIH-Bethesda MD-USA e.mail: linehanm@mail.nih.gov

Regards
G. Pomara MD
(on behalf of all Authors)